Author’s response to reviews

Title: Health care informational challenges for women diagnosed with cervical intraepithelial neoplasia: A qualitative study

Authors:
Carla Freijomil-Vázquez (carla.freijomil@udc.es)
Denise Gastaldo (denise.gastaldo@utoronto.ca)
Carmen Coronado (carmen.coronado@udc.es)
María-Jesús Movilla-Fernández (maria.jesus.movilla@udc.es)

Version: 2 Date: 20 Aug 2019

Author’s response to reviews:

Rosie Brooks
Assistant Editor
BMC Women’s Health
August 20, 2019

Dear Mrs. Rosie Brooks,

We are very grateful for your editorial comments and the reviewers’ comments regarding our manuscript.

Please find enclosed the original article entitled “Health care informational challenges for women diagnosed with cervical intraepithelial neoplasia: A qualitative study”.

A detailed account of the changes made are as follows:

1. Technical comments:
• 1. “In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.” (See page 15)

• 2. “Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.” (See page 16)

• 3. “At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions.” (Entire manuscript)

2. Reviewer 1 comments:

• The comments of reviewer 1 are not consistent with the methodology used in our study (qualitative research). After contacting the editorial board of the journal, the Executive Editor has been decided to dismiss the reviewer 1 comments for not being appropriate (see attached email).

3. Reviewer 2 comments:

• “Accept without revision”, “no additional comments”.

4. English:

• For increase the quality of our paper, our manuscript was edited for proper English language, grammar, punctuation, spelling, and overall style by one or more of the highly qualified native English speaking editors at American Journal Experts (AJE) (see AJE certificate attached).

Thank you again for your recommendations and for considering our re-submission.

Sincerely,

Carla Freijomil-Vázquez
BSc(N), MSc, PhD(C)
Universidade da Coruña

Authors and affiliations:

Carla Freijomil-Vázquez (1,2), Denise Gastaldo (3,4), Carmen Coronado (1,2), María-Jesús Movilla-Fernández (1,2)

(1)Faculty of Nursing and Podiatry, University of A Coruña, Ferrol, Spain
(2)Cardiovascular Research Group (GRINCA), University of A Coruña, Coruña, Spain
(3)Bloomberg Faculty of Nursing, University of Toronto, Toronto, Canada.
(4)Centre for Critical Qualitative Health Research, University of Toronto, Toronto, Canada.